Author’s response to reviews

Title: Iodine level concentration, coverage of adequately iodized salt consumption and factors affecting proper iodized salt utilization among households in North Ethiopia: A Community Based Cross sectional Study

Authors:

Abraham Desta (abaregaydesta@gmail.com)

Usha Kulkarni (ushakulkarni@hotmail.com)

Kidan Abrha (kidanabrha@yahoo.com)

Solomon Worku (solomonw@tutape.org)

Berhe Sahle (berheph@gmail.com)

Version: 3 Date: 27 Mar 2019

Author’s response to reviews:

Dear Brock Madsen,

Editorial office of BMC Nutrition,

Thank you so much for your valuable comments to maintain the manuscript. As per your and reviewers suggestion I kindly incorporated the corrections. The text displayed overlapped in discussion section, line 330- 333 and line 348- 350 page 16 was paraphrased without losing it’s idea. A similar task was also in the section of Ethics approval and consent to participate in the study in line 383- 384 page 18. All the manuscript sections were written according to journals guideline.

//With best regards//

Abraham Aregay Desta

Tigray Health Research Institute, Ethiopia